Author's response to reviews

Title: First multicenter study on multidrug resistant bacteria carriage in Chinese ICUs

Authors:

Xiaojun MA (drmaxiaojun@gmail.com)
Yinghong WU (Wuyinghong0659@163.com)
Liuyi LI (lucyliuyi@263.net)
Qian XU (xqijian356@sohu.com)
Bijie HU (hubijie@vip.sina.com)
Yuxing NI (yuxing_ni@126.com)
Anhua WU (dr_wuanhua@sina.com)
Shumei SUN (sunshumei99@126.com)
Vincent JARLIER (vincent.jarlier@psl.aphp.fr)
Jerome ROBERT (jerome.robert@psl.aphp.fr)

Version: 2
Date: 21 March 2015

Author's response to reviews:

Sir
We have complied with the requests
- line numbering
- competing interest
- abbreviations' list
- emails on the title page
- regarding ethics, we have obtained clearance from all sites before starting the study and this was added in the methods section.

Best regards
J Robert